

Author's response to reviews

Title: Antitumour responses induced by a cell-based Reovirus vaccine in murine lung and melanoma models

Authors:

Ciorsdan Champion (ciorsdan.champion@ucc.ie)

Declan Soden (d.soden@ucc.ie)

Patrick Forde (p.forde@ucc.ie)

Version: 2 Date: 13 Jun 2016

Author's response to reviews:

The Authors made every effort to adequately answer all comments raised by the two reviewers. As the prophylactic and therapeutic protocols protocol cell based vaccinations are very important for this manuscript as they represent key points of the study, I suggest, if possible, to include a panel/figure in which these two protocols are schematically depicted/synthesized.

A schematic has been included giving an overview of the regimes employed for both prophylaxis and therapeutic approaches. This is included in the manuscript as Fig. 1.